Author's response to reviews

Title: Overexpression of GOLPH3 is associated with poor prognosis and clinical progression in pancreatic ductal adenocarcinoma

Authors:

Luan-Jing Zhang (zhangluanjing@126.com)
Ke-Bing Wang (wkb001788@163.com)
Long-shan Liu (iulongshan2003@163.com)
Lian-zhou Chen (chenlz@126.com)
Bao-Gang Peng (pengbaogang@medmail.com.cn)
Li-Jian Liang (lianqlj@medmail.com.cn)
Zhi Li (lizhi@mail.sysu.edu.cn)
Ling Xue (xuel@mail.sysu.edu.cn)
Wen Li (wenli28@163.com)
Jing-Tang Xia (13xia@163.com)

Version: 4
Date: 25 July 2014

Author's response to reviews: see over
Dear Editors,

Thank you for reviewing our manuscript entitled “Overexpression of GOLPH3 is associated with poor prognosis and clinical progression in pancreatic ductal adenocarcinoma”.

We are now giving point-to-point responses to the critiques from editor and re-submitting a revised manuscript now.

According to the journal template, our files are correctly formatted to conform the journal style. I hope that you will find that the current version of our manuscript is suitable for publication in BMC cancer.

If you have any question regarding the re-submitted manuscript, please do not hesitate to contact me.

Sincerely yours,

Wen li, M.D, Ph.D.
Laboratory of General Surgery, The First Affiliated Hospital, Sun Yat-sen University, #58 Zhongshan er Road, Guangzhou, Guangdong 510080, China

Jing-Tang Xia, M.D., Ph.D.
Department of General Surgery, The third Affiliated Hospital, Guangzhou Medical University, Guangzhou, Guangdong 510150, China

**point-to-point responses**

Editorial Request:

Please could you correct the grammatical error on line 132 - "Ehical" should be 'Ethical'.

Reply: We have corrected the grammatical error on line 132 - "Ehical" was changed to be 'Ethical' on line 118 in the revised manuscript.